Author's response to reviews

Title: Sexual behaviors and their correlates among young people in Mauritius: a cross-sectional study

Authors:

Yumiko H Nishimura (yumiko@pbh.med.kyoto-u.ac.jp)
Masako Ono-Kihara (okmasako@pbh.med.kyoto-u.ac.jp)
Jagdish C Mohith (mihealth@intnet.mu)
Renaud NgManSun (renaud.ngmansun@coi-ioc.org)
Takayuki Homma (thomma@p.kanazawa-u.ac.jp)
Ralph J DiClemente (rdiclem@sph.emory.edu)
Delia L Lang (dlang2@sph.emory.edu)
Masahiro Kihara (poghse@pbh.med.kyoto-u.ac.jp)

Version: 4 Date: 28 September 2007

Author's response to reviews: see over
28 September 2007

The Editor
BMC International Health and Human Rights

Re: MS: 1760568572138689 - Sexual behaviors and their correlates among young people in Mauritius: a cross-sectional study”

Dear Editor,

We are delighted to hear that our manuscript will be accepted for publication in BMC International Health and Human Rights. We checked over the manuscript and made some typographical changes as indicated below.

Page 20 second paragraph: PWHA → PLWHA
Page 25 ref No.15: Behavioral Surveillance Surveys → behavioral surveillance surveys
Page 26 ref No.24: HIV/AIDS Survey Indicators Database → HIV/AIDS survey indicators database
Page 29 Table2: Always use a condom → Always used a condom
Page 29 Table2 footnote.c.: people who had had sex → people who had sex
Page 30 Table3 Female, Bivariate, under $P: \chi^2 \rightarrow (\chi^2)^b$

Thank you.

Yours sincerely,

Yumiko H Nishimura
Corresponding author